Author’s response to reviews

Title: Older multimorbid patients’ experiences on integration of services: A systematic review

Authors:
Lilian Boye (Lilian.keene.boye@rsyd.dk)
Christian Mogensen (Christian.backer.mogensen@rsyd.dk)
Tine Mechlenborg (TMechlenborgKristiansen@Gigtforeningen.dk)
Frans Waldorff (fwaldorff@health.sdu.dk)
Pernille Andersen (ptandersen@health.sdu.dk)

Version: 3 Date: 11 Oct 2019

Author’s response to reviews:

The authors thank the reviewers and Editor for the potentially acceptable for publication in BMC Health Services Research.

We thank the Editor for the relevant comments for the manuscript. We hope that we have adequately dealt with the revisions and that the manuscript can now be accepted for publication. The points correspond to the comments.

Best regards
Lilian Keene Boye

Editor Comments:

1. Abbreviations

Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

• We thank the Editor for this comment; we have now removed the author’s abbreviations in the manuscript and added the other used abbreviations in the List of Abbreviations before the declarations section. We also made sure they are defined in the manuscript and table 1. QSR international is the company name of the providers of Nvivo and therefore it is not possible to elaborate this abbreviation further.
2. Clean manuscript

At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.

- Thanks for this comment; the manuscript and table is submitted in a clean version.